Author’s response to reviews

Title: SHORT AND MID-TERM OUTCOMES AND FUNCTIONAL RESULTS IN METALON- METAL HIP RESURFACING ARTHROPLASTY AT 5 YEARS FOLLOW-UP: THE SPANISH EXPERIENCE

Authors:
Olga Susana Pérez-Moro (opemo@hotmail.com)
Marcos Edgar Fernández-Cuadros (marcosefc@hotmail.com)
Inmaculada Neira-Borrajo (Inmaneira@icloud.com)
Eduvigis Aranda-Izquierdo (Eduvigisaranda@gmail.com)
Maria Jesus Albaladejo-Florin (mjflorin@gmail.com)
Rafael Llopis-Miró (rafa@drllopis.com)

Version: 3 Date: 14 Feb 2019

Author’s response to reviews:

Editor Comments have been addressed as follows:

The introduction section has been renamed as Background
The material and methods section has been renamed as Methods

A statement regarding the collection of written informed consent has been included in the "Ethics and Consent to Participate" paragraph in the Declarations section.

Reviewer 2 comments have been addressed as follows:

We changed the wording "friction materials" to "bearing materials" and "friction pairs" to "bearing couples".